Research Paper



Somayeh Haji-Jafari' 💿, Seyed Hossein Ahmadi Tafti² 💿, Ismail Azizi-Fini³ 💿, Mahboubeh Rezaei³ 💿, Fatemeh Atoofª 💿

- 1. Trauma Nursing Research Center, Kashan University of Medical Sciences, Kashan, Iran.
- 2. Chief Surgeon & Director of Thoracic and Cardiac Surgery, Research Center for Advanced Technologies in Cardiovascular Medicine, Tehran Heart Center, Tehran University of Medical Sciences, Iran.
- 3. Trauma Nursing Research Center, Kashan University of Medical Sciences, Kashan, Iran.

4. Department of Biostatistics, Faculty of Health, Kashan University of Medical Sciences, Kashan, Iran.



Citation Haji-Jafari, S., Ahmadi Tafti, S, H., Azizi-Fini, I., Rezaei, M., Atoof, F. 2023. The Effect of Rewarming on the Cognitive Status of Patients After Open-heart Surgery: A Randomized Clinical Trial. *Journal of Client-Centered Nursing Care*, 9(1), pp. 25-34. https://doi.org/10.32598/JCCNC.9.1.305.5

doj https://doi.org/10.32598/JCCNC.9.1.305.5



Article info: Received: 06 Sep 2022 Accepted: 15 Nov 2022 Published: 01 Feb 2023

Keywords:

Rewarming, Hypothermia, Cognitive status, Postoperative complications, Thoracic surgery

ABSTRACT

Background: Hypothermia is usually used during cardiac surgery to further protect against cardiac and cerebral ischemia. However, cognitive impairment could be a common complication after open heart surgery. This study aims to determine the effect of rewarming on the cognitive status of patients after open heart surgery.

Methods: The present study was conducted as a double-blind, randomized clinical trial at Tehran Heart Center, Tehran, Iran. In this clinical trial, 80 patients referred for open heart surgery were selected and randomly assigned into the intervention (n=40) and control (n=40) groups. In the intervention group, rewarming the mattress was started after the patient entered the intensive care unit. Warming continued until the patient's body temperature reached the normothermia level (37°C to 37.5°C). The data were collected by demographic-clinical questionnaire and Mini-Mental State Examination at three time points: before, 7 days, and 1 month after the surgery. The data were analyzed in SPSS software, version 16 using descriptive statistics, independent t test, chi-square test, repeated measure ANOVA, and paired t test. The significance level was considered less than 0.05.

Results: The difference between the Mean±SD scores of cognitive status in the intervention group before the surgery (28.73±0.87), 7 days (27.63±1.03), and 1 month after the surgery (28.93±1.21) was significant (P=0.008). The difference between the Mean±SD scores of cognitive status in the control group before the surgery (28.42±1.02), 7 days (26.61±0.86), and 1 month after the surgery (27.85±1.06) was also significant (P=0.042). The results indicated that the mean score of cognitive status improves after 1 month, and it is greater in the intervention group (F=20.37, df=2, P<0.001). Significant differences were observed between the first and second time of measurements (P<0.0001) and also the second and third time of measurements (P<0.0001).

Conclusion: Rewarming patients after open heart surgery improves their cognitive status. Therefore, this procedure can be used as a safe and non-invasive method to prevent cognitive complications by nurses after open heart surgery.

.....

* Corresponding Author: Ismail Azizi-Fini, PhD. Address: Trauma Nursing Research Center, Kashan University of Medical Sciences, Kashan, Iran. Tel: +98 (912) 2475747 E-mail: azizifinies@yahoo.com

Highlights

• Cardiovascular diseases are one of the most important causes of death worldwide.

• Open heart surgery is one of the main treatments for coronary artery occlusion, which leads to complications such as cognitive impairment in patients.

• Hypothermia complications are among the factors that cause cognitive impairment in patients undergoing openheart surgery.

• The results showed that rewarming the patient after open heart surgery can neutralize the effect of preoperative hypothermia on the patient's cognitive status.

Plain Language Summary

Open heart surgery is a standard treatment for coronary artery obstruction. The nature of the surgery and the use of hypothermia during the surgery can cause many complications in the long run, including cognitive impairment. Warming up the body after surgery is initially done in the operating room by a cardiopulmonary bypass machine. But after entering the ward, the patients suffer hypothermia again, which may affect their cognitive status. The results indicated that rewarming the patient after open heart surgery can neutralize the effect of preoperative hypothermia on the patient's cognitive status.

1. Introduction

ardiovascular disease (CVD) is the leading cause of mortality and morbidity worldwide. Approximately 17.9 million global deaths from CVD were reported in 2015. Ischemic Heart Disease (IHD) and

stroke are the top two leading causes of CVD worldwide, and more than 22.2 million individuals will die from CVD by 2030 (Pepera et al., 2022).

The mortality rate from non-infectious diseases such as cardiovascular problems has risen from 66.17% to 68.98% in the Iranian population from 2006 to 2016 (Kazemi et al., 2021). According to other studies in Iran, cardiovascular diseases account for nearly half of all deaths (Rezaei et al., 2022; Sadeghi et al., 2017). Coronary artery obstruction is the leading cause of death among these cardiovascular problems (Izadi-Avanji et al., 2022). Complications of infarction or heart attack cause problems for many patients (Poorzand et al., 2019).

Open heart surgery is one of the most common treatments for patients who have suffered a heart attack. However, like any other surgery, an open heart surgery can have early and late complications. These include hemodynamic instability, infection, cardiac tamponade, non-fusion of the sternum bone, and cognitive impairment (Maghami et al., 2020; Afazel et al., 2017). Cogni-

tive impairment after open heart surgery affects the patient's quality of life (Li et al., 2022; Martirosyan et al., 2020). Despite the progress in heart surgery techniques and effective brain protection strategies, cognitive dysfunction is still significant. Cognitive dysfunction after surgery has been defined as a decrease in a person's cognitive function after surgery and anesthesia compared to the initial level before surgery (Tan and Amoako, 2013). It is more common in patients undergoing heart surgery than in those undergoing other types of surgery. Postoperative cognitive dysfunction (POCD), characterized by impaired attention, concentration, and memory with possible long-term consequences, is a frequent neurological sequela after cardiac surgery. POCD can be classified into two types based on its duration: short-term and long-term. A transient cognitive decline usually lasts up to 6 weeks after a cardiac procedure, affecting 20%-50% of patients, while a subtle decline in cognitive function occurs 6 months after a cardiac procedure, affecting 10%-30% of patients (Bruggemans, 2013; Tan and Amoako, 2013). The occurrence of this disorder after surgery has been reported to be 45%-88% at the time of hospital discharge and 15%-36% six weeks after discharge so the patient may suffer from this disorder for even up to 5 years after surgery (Yang et al., 2022; Bruggemans, 2013). Cognitive dysfunction after surgery leads to frequent visits of patients to medical centers and a decrease in their quality of life, leading to an increase in the health burden. In addition, this condition puts a significant economic burden on the health system and the families of these patients, which is estimated to be 18 billion dollars annually (Sakusic et al., 2018).

Hypothermia during surgery is one of the most important causes of cognitive failure after heart surgery. The duration of hypothermia in surgery plays a major role in determining the neurologic outcomes of patients, but the extent of this effect is still debated (Patel et al., 2015). Since there is no known treatment for cognitive dysfunction after open heart surgery, emphasis is placed on implementing non-invasive nursing intervention strategies, including rewarming to reduce postoperative central nervous system damage (Berger et al., 2018). It is widely believed that rewarming after deep hypothermia induced by a cardiopulmonary bypass (CPB) machine is favorable for rapid patient recovery (Gocoł et al., 2021).

Various studies have been conducted on warming patients, which have entirely focused on the issue of warming during open heart surgery (Saad and Aladawy, 2013; Currey and Botti, 2005). Rewarming has been shown to improve cognitive function (Patel et al., 2015). Some studies have found it helpful to perform non-invasive interventions such as warming the skin during surgery with a hot water mattress or hot compressed air. Some also point to warming the face, knees, and shoulders of patients after surgery as a simple and effective method and consider it effective in preventing the temperature from dropping again (Rajagopalan et al., 2008; Bezerra et al., 2021; Ciğerci et al., 2020).

Various studies have been conducted on the effect of temperature with different degrees during surgery on the cognitive function of patients after surgery. However, studies do not reveal a clear difference between the induction of mild hypothermia or normothermia and their effect on neurocognitive function at discharge and three months later. The results of two studies comparing the effects of normothermia and hypothermia during cardiac surgery on cognitive function after surgery showed a significant improvement in cognitive function in the normothermia group (Shaaban-Ali et al., 2002; Grimm et al., 2000). The results of two other studies comparing the effect of hypothermia and normothermia during heart surgery show that cognitive status improves in the hypothermia group (Kadoi and Goto, 2006; Boodhwani et al., 2006). Meanwhile, the results of some other studies comparing the effect of hypothermia and normothermia on the nervous system function after heart surgery show no difference between the use of these two methods (Grigore et al., 2009; Reinsfelt et al., 2012).

Because most studies focus on rewarming during surgery and few on rewarming after surgery, as well as the conflict between the results of the studies, the present study aims to determine the effect of rewarming on the cognitive status of patients after open heart surgery.

2. Materials and Methods

Study setting and sample

This double-blind, randomized clinical trial was conducted from September to December 2020 at Tehran Heart Center in Tehran-Iran. The initial sample of the study was determined through consecutive sampling. A total of 80 subjects were selected based on the inclusion criteria and randomly allocated into two groups of 40 by generating ten blocks of four using online blocked randomization software (Sealed Envelope Ltd., 2022). In the online randomization site, letter A was assigned to the intervention group and B to the control group. The patients were unaware of being placed in the intervention or control group. In addition, the statistical analyst did not know the names of the groups. The sample size was determined based on mean arterial blood pressure from the study of Murphy (Murphy et al., 2009), in which the Mean±SD arterial blood pressure was 13.5±1.9 and changed to 15.2±0.7 mmHg. Then, with a type I error of 0.05, a type II error of 0.2, an S1 of 1.9, an S2 of 0.7, a µ1 of 13.5, and a µ2 of 15.2, the sample size for each study group was estimated to be 37. However, due to possible attrition, the sample size in each group was increased to 40 (Equation 1).

1.
$$n = \frac{2(z_{1-\alpha^2} + z_{1-\beta})^2(s_1 + s_2)^2}{(\mu_1 + \mu_2)^2} = 37$$

The potential subjects were elective patients who were candidates for open heart surgery. Other inclusion criteria were having a Mini-Mental State Exam (MMSE) score higher than 24 and not having an intra-aortic balloon before surgery. The exclusion criteria were prolongation of intubation time (more than 12 hours) due to respiratory reasons, resternotomy due to postoperative surgical reasons, having an intra-aortic balloon after surgery, and re-intubation after surgery, patients undergoing reoperation, aortic aneurysm or dissection, death of the patient, return of the patient to the operating room for any reason, and prolongation of surgery for any reason more than five hours. The outcome measure in this study is the changes in mean cognitive function in different time intervals.

Study procedure

As part of a routine cardiac operating room procedure, all patients become hypothermic during cardiac surgery, with their body temperature lowered to 25°C-32°C. Immediately after surgery, the body temperature gradually increases to 36°C, which is the first step to warming the patient in the operating room. All patients undergoing open heart surgery are required to undergo this procedure.

Before the patient entered the intensive care unit, an electric warming mattress was placed on the patient's bed. It was model EU133, the electrical under blanket and heating pad-EmsiG GmbH. The upper surface is made of small woolen fibers, and the lower surface of polyester (dimensions: 80 x 150 cm, power: 75 W, the temperature range: 35°C to 50°C). The blanket performance was examined, and its temperature was set at 37.5°C. The bed was sent to the operating room, and the patient was transferred from the operating room bed to the bed with the electric mattress so that the mattress was placed under the patient from the shoulder to the ankle. This way, rewarming was started immediately after the surgery, and the patient's body temperature was gradually raised to 36°C. Although it was the first step of rewarming and the same for all patients, we did not consider it part of the intervention, and the rewarming time was calculated from the entry of the patients to the ICU till their body temperature reached preoperative normothermic condition (i.e., between 37°C and 37.5°C).

This way, all patients in the intervention group were rewarmed using an electric mattress, but the control group received routine care.

Study instruments

The data were collected using demographic information form (age, gender, education, marital status, occupation, smoking), surgical information form (type of surgery, duration of surgery, duration of the aortic clamp, number of grafts, and duration of intubation), the MMSE, and an axillary temperature probe. MMSE has been translated into different languages and standardized in different cultures. This 30-point test measures different cognitive functions and provides a general assessment of the subject's cognitive status. It is an 11-question measure that tests 5 areas of cognitive function: orientation, registration, attention and calculation, recall, and language. The scoring procedure in this questionnaire is as follows: orientation=10 points, memory=6 points (including recording 3 points and recalling 3 points); attention and calculation=5 points; assessment of language abilities and understanding=8 points. Also, 1 score is given to visual and spatial abilities. The minimum and maximum scores for the MMSE are 0 and 30, respectively. Scores above 24 indicate no cognitive problem, scores between 20 and 23 indicate mild cognitive impairment, scores 10 to 19 indicate moderate impairment, and 0 to 9 indicate severe impairment (Seyedian et al., 2007).

To measure body temperature, an axillary temperature probe connected to a monitor continuously measured and monitored the body temperature of the patients. Before use, the probes were calibrated by a medical equipment engineer. To ensure the accuracy of the thermometer, the measurement results were compared with those of a mercury thermometer for one patient, and the results were identical.

The cognitive status of patients was measured in three time intervals; before the surgery, 7 days, and 1 month after the surgery (Cormack et al., 2012; Nathan et al., 2001). In the third time interval (1 month after the surgery), the patients completed the questionnaires when they returned to the hospital for their cardiac rehabilitation.

The data were entered into SPSS software, version 16 (SPSS Inc., Chicago, IL, USA) and analyzed using descriptive statistics, independent t test, and paired t test. The Kolmogorov-Smirnov test was used to check if the data had a normal distribution. The P value was considered less than 0.05.

3. Results

A total of 100 patients were consecutively assessed for eligibility, of which 8 patients were excluded due to unwillingness to participate in the study, and 12 patients due to lack of eligibility criteria (Figure 1).

The results showed that the Mean±SD ages of the intervention and control groups were 58.71 ± 13.8 and 57.94 ± 9.7 years, respectively. The Mean±SD durations of intubation in the intervention and control groups were 11.33 ± 5.19 and 12.48 ± 5.7 hours, respectively. Most of the subjects in the intervention (65.72%) and control (66.66%) groups were men; the statistical tests did not show any statistically significant difference between the two groups in terms of individual characteristics, including age, sex, type of surgery, tobacco and substance abuse in the past year, underlying diseases, number of grafts, tracheal tube time, CPB time, aortic clamping time, and duration of surgery (P>0.05) (Table 1).

Variables		No. (%)/Mean±SD		_
		Intervention (n=35)	Control (n=36)	P
Sex	Male	23 (65.72)	24(66.66)	0.14
	Female	12 (34.28)	12(33.33)	0.14
Tobacco and substance abuse in the past year	Yes	35 (100)	36(100)	1
	No	0	0	
Type of surgery	CABG	25 (71.42)	23(63.88)	
	Valve replacement	8 (22.85)	12(33.33)	0.43
	Both	2 (5.71)	1(2.77)	
Underlying diseases	HTN	10 (28.6)	8(22.2)	
	Diabetes	6(17.1)	9(25)	0.71
	HTN & Diabetes	15(42.9)	13(36.1)	
	No	4 (11.4)	6(16.7)	
Age (y)		58.71±13.8	57.94±9.7	0.78
Number of grafts		2.4±0.20	2.1±0.12	0.54
Duration of surgery (h)		1.37±0.86	1.38±0.54	0.90
Tracheal tube time (d)		11.33±5.19	12.48±5.7	0.37
CPB time (min)		68.40±35.4	70.08±19.6	0.80
Aortic clamping time (min)		37.68±20.12	44.11±14.90	0.13

Table 1. Demographic and clinical characteristics of the patients in the intervention and control groups

CABG: Coronary artery bypass graft; HTN: Hypertension; CPB: Cardiopulmonary bypass.

In the repeated measures analysis, Mauchly's test illustrated that sphericity was assumed (P>0.05); then, the degrees of freedom were corrected using the sphericity assumed test. The analysis of variance with repeated measures showed significant changes in cognitive status over time (F=20.37, P=0.000). Also, a significant difference was observed between the mean scores of the cognitive status of the two groups (F=18.12, P=0.000). However, the changes in cognitive status at three time points between the two groups were not significant (F=1.12, P=0.330). Moreover, the analysis of variance with repeated measures performed separately for each group showed that the mean scores of cognitive status changed significantly over time in the intervention (F=5.55, P=0.008) and control (F=4.70, P=0.000) groups. The results of the independent t test at three time points showed that before surgery, there was no difference between the two groups in terms of cognitive status, but at the time points of 7 days and 1 month after the surgery, there was a significant difference between the control and intervention groups (P<0.05) (Table 2).

4. Discussion

The results showed that rewarming after cardiac surgery significantly improved the cognitive status of patients 7 days and 1 month after the intervention. Cognitive disorders after heart surgery may occur due to transient hypoxia, cerebral hypoxia, long duration of the aortic clamp, hemoglobin drop during surgery, severe or prolonged drop in arterial blood pressure, and a drop in body temperature during the CPB (Rezamasouleh et al., 2014). Patel et al.'s research showed that slow rewarming after heart surgery could improve cognitive performance (Patel et al., 2015). The results of his research are consistent with the present study. It has been shown that rewarming after heart surgery reduces cognitive impairment and increases memory and attention in patients 1

Group		Mea	an±SD	
Tim		Control (n=35)	Intervention (n=36)	- Independent t-test
24 hours before the surgery (T1)		28.42±1.02	28.73±0.87	t=1.02, P=0.316
Seven days after the surgery (T2)		26.61±0.86	27.63±1.03	t=3.41, P=0.002
One month after the surgery (T3)		27.85±1.06	28.93±1.21	t=2.15, P=0.038
Time effect in each group		F=4.70, P=0.000	F=5.55, P=0.008	-
Results of repeated measure ANOVA	Within groups		F=20.37, P=0.000	-
	Between groups		F=18.12, P=0.000	-
	Groups×time		F=1.12, P=0.330	-
				Client- Centered Nursing Care

Table 2. Comparing the mean scores of cognitive status before, 7 days, and 1 month after the heart surgery in the intervention and control groups

week (Cormack et al., 2012; Nathan et al., 2001) and 3 months after surgery (Polderman, 2008; Linassi et al., 2022; Cormack et al., 2012). Also, it is shown in a study that rewarming after coronary artery bypass surgery improves cognitive function in rats (De Lange et al., 2008). In the mentioned study, rats underwent surgery for 90

It has been reported in some studies that surgery using extra-corporeal blood circulation (extra-corporeal)

has been found to cause short-term and long-term cogni-

minutes, and their cognitive function was measured

based on brain function and cerebral perfusion.





Client- Centered Nursing Care

tive dysfunction in 55% of cases (Chernov et al., 2006; Belrose and Noppens, 2019). Therefore, it seems that short-term and long-term complications of cognitive impairment after open heart surgery can be prevented by performing interventions such as rewarming. It is believed that rewarming increases oxygen supply to the brain and improves cognitive status. (Ali et al., 2014; Mohmaed Shaaban et al., 2014; Hu et al., 2016; Sugiura et al., 2021). It has been shown that a CPB machine alone can result in neurological damage. There are several reasons why this complication occurs, including the non-pulsatile blood supply caused by the bypass device, the activation of complements and inflammatory mediators, the activation of platelets, and their accumulation as microemboli (Puthettu et al., 2021).

The prevalence of psychological problems after coronary artery bypass surgery has been emphasized (Burkauskas et al., 2018). Among the cognitive disorders after heart surgery is disorientation to time and place, and disorder in calculating or drawing shapes. Meanwhile, providing oxygen and warming the patient can be effective in solving the neurological disorders of patients after open heart surgery (Burkauskas et al., 2018; Ottens et al., 2017). The findings of these studies are consistent with the present research. In the current study, the MMSE was used to measure the cognitive status of the patients after surgery. This questionnaire includes dimensions of orientation to time and place, performing calculations and drawing shapes, and remembering events. The results showed that patients had disorders in these fields after open heart surgery. However, 1 month after the intervention, these disorders decreased. So that after the intervention, the score of cognitive status in the intervention group increased significantly compared to the control group. Contrary to the present study, Engelman et al. showed no relationship between cognitive status and hyperthermia after open heart surgery. He suggested more studies in this field (Engelman et al., 2019). The results of his research are not consistent with the present study. The inconsistency may be related to the method of warming the patient, the speed of warming, and the tool for measuring the cognitive status. It may also be due to the difference between the measurement time points.

One of the limitations of this study is the possible effects of uncontrollable confounding variables. Another limitation was the impossibility of choosing a specific surgeon for the studied patients, which may have affected the conditions of the samples. Also, hemodynamic changes after cardiac surgery may have affected the results, which were beyond the control of the researchers.

5. Conclusion

The study showed that rewarming patients after open heart surgery improve their cognitive status. Therefore, it can be used as a safe and non-invasive method to enhance the cognitive status of patients after heart surgery by nurses.

Ethical Considerations

Compliance with ethical guidelines

The Ethics Committee of Kashan University of Medical Sciences approved the study (Code: IR.KAUMS. NUHEPM.REC.1398.016). It was also registered in the Iranian Registry of Clinical Trials (IRCT ID: IRCT20100124003146N8). All subjects were informed of the intervention and signed the written informed consent form. They were also assured of data confidentiality and informed that they could withdraw from the study at any time without affecting their treatment. Necessary permissions were also sought from the officials at Tehran Heart Center. We also observed all patients' rights according to the Helsinki Declaration. The subjects were also assured that attending the study would not cost them anything.

Funding

This article was extracted from a Master's thesis Somayeh Haji-Jafari the Research Deputy of Kashan University of Medical Sciences for their financial support (Grant No: 98051).

Authors' contributions

Conceptualization and data curation: Ismail Azizi-Fini and Somayeh Haji-Jafari; Data analysis and software: Fatemeh Atoof; Funding acquisition, resources, and project administration: Ismail Azizi-Fini; Investigation: Ismail Azizi-Fini, Somayeh Haji-Jafari, and Mahboubeh Rezaei; Methodology: Ismail Azizi-Fini and Mahboubeh Rezaei; Supervision and validation, writing, review, and editing: Ismail Azizi-Fini and Seyed Hossein Ahmadi Tafti; Writing the original draft: Ismail Azizi-Fini and Somayeh Haji-Jafari.

Conflict of interest

The authors declared no conflict of interest.

References

- Afazel, M. R., et al., 2017. The effects of continuous positive airway pressure mask on hemodynamic parameters after open heart surgery: A randomized controlled trial. *Nursing and Midwifery Studies*, 6(3), pp. 109-14. [DOI:10.4103/nms.nms_20_17]
- ALI, M. S., et al., 2014. Effect of slow versus rapid rewarming on jugular bulb oxygen saturation in adult patients undergoing open heart surgery. *Saudi Journal of Anaesthesia*, 8 (2), pp. 178-82. [DOI:10.4103/1658-354X.130698] [PMID] [PMCID]
- Belrose, J. C. & Noppens, R. R., 2019. Anesthesiology and cognitive impairment: A narrative review of current clinical literature. *BMC Anesthesiology*, 19(1), pp. 241. [DOI:10.1186/s12871-019-0903-7] [PMID] [PMCID]
- Berger, M., et al., 2018. Neurocognitive function after cardiac surgery: From phenotypes to mechanisms. *Anesthesiology*, 129(4), pp. 829-51. [DOI:10.1097/ALN.00000000002194] [PMID] [PMCID]
- Bezerra, A. S. M., et al., 2021. Effect of nurse-initiated forced-air warming blanket on the reduction of hypothermia complications following coronary artery bypass grafting: A randomized clinical trial. *European Journal of Cardiovascular Nursing*, 20(5), pp. 445-53. [DOI:10.1093/eurjcn/zvaa023] [PMID]
- Boodhwani, M., et al., 2006. Predictors of early neurocognitive deficits in low-risk patients undergoing on-pump coronary artery bypass surgery. *Circulation*,114(1 Suppl), pp. 461-6. [DOI:10.1161/CIRCULATIONAHA.105.001354]
- Bruggemans E. F., 2013. Cognitive dysfunction after cardiac surgery: Pathophysiological mechanisms and preventive strategies. *Netherlands Heart Journal*, 21(2), pp. 70-3. [DOI:10.1007/ s12471-012-0347-x] [PMID] [PMCID]
- Burkauskas, J., et al., 2018. Cognitive function in patients with coronary artery disease: A literature review. *Jour*nal of International Medical Research, 46(10), pp. 4019-31. [DOI:10.1177/0300060517751452] [PMID] [PMCID]
- Chernov, V. I., et al., 2006. Short-term and long-term cognitive function and cerebral perfusion in off-pump and on-pump coronary artery bypass patients. *European Journal of Cardio-thoracic Surgery*, 29(1), pp. 74-81. [DOI:10.1016/j.ejcts.2005.10.001] [PMID]
- Cigerci, Y., et al., 2020. Comparison of body temperature, normothermia, and extubation times of patients heated with forced air warming method based on whether patients underwent on-pump or off-pump coronary artery bypass graft. *Florence Nightingale Journal of Nursing*, 28(1), pp. 33-40. [PMID] [PMCID]
- Cormack, F., et al., 2012. A meta-analysis of cognitive outcome following coronary artery bypass graft surgery. *Neuroscience* & Biobehavioral Reviews, 36(9), pp. 2118-29. [DOI:10.1016/j. neubiorev.2012.06.002] [PMID]
- Currey, J. & Botti, M., 2005. The haemodynamic status of cardiac surgical patients in the initial 2-h recovery period. *European Journal of Cardiovascular Nursing*, 4(3), pp. 207-14. [DOI:10.1016/j.ejcnurse.2005.03.007] [PMID]
- De Lange, F., et al., 2008. The effect of limited rewarming and postoperative hypothermia on cognitive function in a rat cardiopulmonary bypass model. *Anesthesia and Analgesia*, 106(3), pp. 739-45. [DOI:10.1213/ane.0b013e318162d026] [PMID]

- Engelman, D. T., et al., 2019. Guidelines for perioperative care in cardiac surgery: Enhanced recovery after surgery society recommendations. *JAMA Surgery*, 154(8), pp. 755-66. [DOI:10.1001/jamasurg.2019.1153] [PMID]
- Gocol, R., et al., 2021. The role of deep hypothermia in cardiac surgery. International Journal of Environmental Research and Public Health, 18(13), pp. 7061. [DOI:10.3390/ijerph18137061] [PMID] [PMCID]
- Grigore, A. M., et al., 2009. A core review of temperature regimens and neuroprotection during cardiopulmonary bypass: Does rewarming rate matter?, *Anesthesia & Analgesia*, 109(6), pp. 1741-51. [PMID]
- Grimm, M., et al., 2000. Normothermic cardiopulmonary bypass is beneficial for cognitive brain function after coronary artery bypass grafting-a prospective randomized trial. *European Journal of Cardio-thoracic Surgery: Official Journal of the European Association for Cardio-thoracic Surgery*, 18(3), pp. 270-5. [DOI:10.1016/S1010-7940(00)00510-8] [PMID]
- Grocott, H. P., et al., 2002. Postoperative hyperthermia is associated with cognitive dysfunction after coronary artery bypass graft surgery. *Stroke*, 33(2), pp. 537-41. [PMID]
- Hu, Z., et al., 2016. Effects of hypothermic cardiopulmonary bypass on internal jugular bulb venous oxygen saturation, cerebral oxygen saturation, and bispectral index in pediatric patients undergoing cardiac surgery: A prospective study. *Medicine*, 95(2), pp. e2483. [DOI:10.1097/ MD.00000000002483] [PMID] [PMCID]
- Izadi-Avanji, F. S., Esmaeli, T. & Sadat, Z., 2022. The role of information and communication technology use in predicting adherence to treatment in patients with coronary artery disease. *Journal of Client-Centered Nursing Care*, 8(2), pp. 111-8. [DOI:10.32598/JCCNC.8.2.280.5]
- Kadoi, Y. & Goto, F., 2006. Factors associated with postoperative cognitive dysfunction in patients undergoing cardiac surgery. *Surgery Today*, 36(12), pp. 1053-7. [PMID]
- Kazemi, E., et al., 2021, Trends in premature mortality in the Islamic Republic of Iran: Probability of dying between ages 30 and 70 years. *Eastern Mediterranean Health Journal*, 27(1), pp. 33-40. [DOI:10.26719/emhj.20.067] [PMID]
- Li, X., Jia, Z. & Yan, Y., 2022. Ticagrelor for prevention of stroke and cognitive impairment in patients with vascular high-risk factors: A meta-analysis of randomized controlled trials. *International Journal of Cardiology*, 353, pp. 96-102. [PMID]
- Linassi, F., et al., 2022, Targeted temperature management in cardiac surgery: A systematic review and meta-analysis on postoperative cognitive outcomes. *British Journal of Anaesthesia*, 128(1), pp. 11-25. [PMID]
- Maghami, M., et al., 2020. The effect of aromatherapy with peppermint essential oil on nausea and vomiting after cardiac surgery: A randomized clinical trial. *Complementary Therapies in Clinical Practice*, 40, pp. 101199. [DOI:10.1016/j. ctcp.2020.101199] [PMID]
- Martirosyan, A. K., et al., 2020. [Early and long-term outcomes of bilateral inthernal thoracic artery grafting for coronary artery bypass surgery (Russian)]. *Khirurgiia (Mosk)*, 3, pp. 74-81. [DOI:10.17116/hirurgia202003174] [PMID]

- Ali, M. S., et al., 2014. Effect of slow versus rapid rewarming on jugular bulb oxygen saturation in adult patients undergoing open heart surgery. *Saudi Journal of Anaesthesia*, 8(2), pp. 178-82. [DOI:10.4103/1658-354X.130698] [PMID] [PMCID]
- Murphy, G. S., Hessel, E. A. & Groom, R. C., 2009. Optimal perfusion during cardiopulmonary bypass: An evidence-based approach. Anesthesia & Analgesia, 108(5), pp. 1394-417. [PMID]
- Nathan, H. J., et al., 2001. Neuroprotective effect of mild hypothermia in patients undergoing coronary artery surgery with cardiopulmonary bypass: A randomized trial. *Circulation*, 104(12 Suppl 1), pp. 185-91. [PMID]
- Ottens, T. H., et al., 2017. Brain volume and cognitive function in patients with revascularized coronary artery disease. *International Journal of Cardiology*, 230, pp. 80-4. [DOI:10.1016/j. ijcard.2016.12.079] [PMID]
- Patel, N., Minhas, J. S. & Chung, E. M. L., 2015. Risk factors associated with cognitive decline after cardiac surgery: A systematic review. *Cardiovascular Psychiatry and Neurology*, 2015, pp. 370612. [DOI:10.1155/2015/370612] [PMID] [PMCID]
- Pepera, G., et al., 2022. Epidemiology, risk factors and prognosis of cardiovascular disease in the coronavirus disease 2019 (COVID-19) pandemic era: A systematic review. *Reviews* in Cardiovascular Medicine, 23(1), pp. 28. [DOI:10.31083/j. rcm2301028] [PMID]
- Polderman, K. H., 2008. Induced hypothermia and fever control for prevention and treatment of neurological injuries. *Lancet*, 371(9628), pp. 1955-69. [DOI:10.1016/S0140-6736(08)60837-5] [PMID]
- Poorzand, H., et al., 2019. Risk factors of premature coronary artery disease in Iran: A systematic review and meta-analysis. European *Journal of Clinical Investigation*, 49(7), pp. e13124. [DOI:10.1111/eci.13124] [PMID]
- Puthettu, M., Vandenberghe, S. & Demertzis, S., 2021. Effect of cannulation site on emboli travel during cardiac surgery. *Journal of Cardiothoracic Surgery*, 16(1), pp. 181. [DOI:10.1186/ s13019-021-01564-1] [PMID] [PMCID]
- Rajagopalan, S., et al., 2008. The effects of mild perioperative hypothermia on blood loss and transfusion requirement. *Anesthesiology*, 108(1), pp. 71-7. [DOI:10.1097/01. anes.0000296719.73450.52] [PMID]
- Reinsfelt, B., et al., 2012, Cerebrospinal fluid markers of brain injury, inflammation, and blood-brain barrier dysfunction in cardiac surgery. *The Annals of Thoracic Surgery*, 94(2), pp. 549-55. [DOI:10.1016/j.athoracsur.2012.04.044] [PMID]
- Rezaei, F., et al., 2022. Estimation of 10-year risk of cardiovascular diseases using who risk prediction charts: A populationbased study in southern Iran. *Iranian Journal of Public Health*, 51(7), pp. 1667-76. [DOI:10.18502/ijph.v51i7.10101]
- Rezamasouleh, S., et al., 2014. [Chest drainage and its associated factors in patients who undergone coronary artery bypass grafting (CABG) surgery and admitted to the intensive care unit (ICU) (Persian)]. *Journal of Holistic Nursing and Midwifery*, 24(3), pp. 10-19. [Link]
- Saad, H., & Aladawy, M., 2013. Temperature management in cardiac surgery. *Global Cardiology Science and Practice*, 2013, (1), pp. 44-62. [PMID] [PMCID]

- Sadeghi, M., et al., 2017. Modeling the burden of cardiovascular diseases in Iran from 2005 to 2025: The impact of demographic changes. *Iranian Journal of Public Health*, 46(4), pp. 506-16. [PMID]
- Sakusic, A., et al., 2018. Potentially modifiable risk factors for long-term cognitive impairment after critical illness: A systematic review. *Mayo Clinic Proceedings*, 93(1), pp. 68-82. [PMID]
- Seyedian, M., et al., 2007. [Validity of the Farsi version of minimental state examination (Persian)]. *Journal of Medical Council* of Islamic Republic of Iran, 25(4), pp. 408-14. [Link]
- Sealed Envelope Ltd., 2022. Create a blocked randomisation list. Retrived from: from: [Link]
- Shaaban-Ali, M., et al., 2002. Changes in serum S100β protein and Mini-Mental State Examination after cold (28° C) and warm (34° C) cardiopulmonary bypass using different blood gas strategies (alpha-stat and pH-stat). Acta Anaesthesiologica Scandinavica, 46(1), pp. 10-16. [DOI:10.1046/j.0001-5172.2001.00000.x] [PMID]
- Sugiura, A., et al., 2021. Effective method of monitoring cerebral tissue oxygen saturation in cardiac surgery patients by combined use of tNIRS-1 and bispectral index. *Scientific Reports*, 11(1), pp. 24126. [DOI:10.1038/s41598-021-03527-x] [PMID] [PMCID]
- Tan, A. M. Y. & Amoako, D., 2013. Postoperative cognitive dysfunction after cardiac surgery. *Continuing Education in Anaesthesia Critical Care & Pain*, 13(6), pp. 218-23. [DOI:10.1093/ bjaceaccp/mkt022]
- Yang, X., et al., 2022. Identification of individuals at risk for postoperative cognitive dysfunction (POCD). *Therapeutic Advances in Neurological Disorders*, 15, pp. 17562864221114356. [DOI:10.1177/17562864221114356] [PMID] [PMCID]

This Page Intentionally Left Blank